# Deploy Bot


| Host                                     | Website                                           | Introduction                                                        |
| :---                                     | :----                                             | ---:                                                                |
| [Digital Ocean ↗](/deploy/digital-ocean) | [digitalocean.com](https://www.digitalocean.com/) | Esay, free trail, Credit Card Requre                                |
| [Heroku ↗](/deploy/heroku)               | [heroku.com](https://www.heroku.com/)             | 100% Deploy Efficiency, 5$ per month `recomend`                     |
| [Koyeb ↗](/deploy/koyeb)                 | [koyeb.com](https://www.koyeb.com/)               | Host 2 Apps free, Esay, Credit Card Requre                          |
| [Railway ↗](/deploy/railway)             | [railway.app](https://www.railway.app/)           | User Friendly, Esay & free trail No Credit Card Requre `recomend`   |
| [Render ↗](/deploy/render)               | [render.com](https://www.render.com/)             | Good Ui & Ux, Free Trail, No Credit Card Requre                     |